Author's response to reviews

Title: Practice patterns of naturopathic physicians: results from a random survey of licensed practitioners in two US States

Authors:

Heather S Boon (heather.boon@utoronto.ca)
Daniel C Cherkin (cherkin.d@ghc.org)
Janet Erro (erro.j@ghc.org)
Karen J Sherman (sherman.k@ghc.org)
Bruce Milliman (doctorwbm@earthlink.net)
Jennifer Booker (jenniferbooker@earthlink.net)
Elaine H Cramer (ecramer@telus.net)
Michael J Smith (Michael.J.Smith@hc-sc.gc.ca)
Richard A Deyo (deyo@u.washington.edu)
David M Eisenberg (david_eisenberg@hms.harvard.edu)

Version: 4 Date: 14 October 2004

Author's response to reviews:

So sorry to have missed these required changes last time. The following formatting changes have been made as requested:

"By" - Please remove this from the title page.
Tables - Unfortunately, we will be unable to display shading within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.